Author's response to reviews

Title: Differences in the genetic pathway(s) between Vietnamese and Japanese colorectal cancers

Authors:

Tomohiro Miwata (tmiwata@hiroshima-u.ac.jp)
Toru Hiyama (tohiyama@hiroshima-u.ac.jp)
Duc T Quach (quachtrongduc2002@yahoo.com)
Huy M Le (leminhhuy@yahoo.com)
Ha NT Hua (hthngocha@yahoo.com)
Shiro Oka (oka4683@hiroshima-u.ac.jp)
Shinji Tanaka (colon@hiroshima-u.ac.jp)
Koji Arihiro (arihiro@hiroshima-u.ac.jp)
Kazuaki Chayama (chayama@hiroshima-u.ac.jp)

Version: 2
Date: 19 June 2014

Author's response to reviews:

Dear Prof. Tonilynn Manibo,

Thank you for your e-mail.

I have revised the manuscript according to your comments.

1. I included line and page numbers.
2. I included the Competing interests section between the Conclusions and Authors' contributions.
3. I included the Authors' contributions section.
4. I included the Acknowledgements section.
5. I provided email address of each co-author in the Title page.

Thank you again for your help.

Best regards,

Toru Hiyama, MD, PhD
Health Service Center
Hiroshima University
Japan